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Activities during the previous 
week

• Continue verification of SMT HDI recovery effort
– Exercise power supplies
– Complete SMT HV tests

• Prepare for and recover from scheduled power outage
– Repair failed calorimeter preamp power supply

• Complete safety system tests
• Installation of AFEII in FPS slots continues

– 23 of 32 boards currently installed
• Continue testing new AFEII data format on platform
• Clear and close the gaps between the calorimeter
• Pump out vacuum jackets
• Low beta quads have been re-aligned
• Close pit access door
• Prepare cathedrals for closing
• Trigger list studies and development continue
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Plans for last week of 
shutdown

• Monday
– Close CFs and EFs
– Search and secure collision hall
– Power up solenoid and toroid tests
– Record cosmic triggers with full detector during evening shift

• Tuesday
– Supervised access to collision hall resumes
– Erect scaffolding in south west corner
– Re-insulate low beta quads
– South B Layer VStar survey

• Wednesday
– Closing survey
– Complete insulation of low beta quads
– Clean south west central muon proportional drift tube high voltage distribution cards

• Thursday
– Take down scaffolding
– Resume scheduling full shift coverage

• Exercise examines and other monitoring tools with all detectors in global run (on new platforms)
• Evaluate SES alarm status and respond accordingly

– Record magnet off cosmic triggers
• Friday

– Verify detector ready for search and secure
– Search and secure collision hall
– Continue exercising DAQ and detector to identify and address remaining issues

• Saturday 
– Run detector as necessary to investigate any remaining issues

• Plan to be ready to make good use of collisions when delivered
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